Author's response to reviews

Title: Prolactinomas, Cushing's disease and acromegaly: debating the role of medical therapy for secretory pituitary adenomas

Authors:

Beverly MK Biller (bbiller@partners.org)
Annamaria Colao (colao@unina.it)
Stephan Petersenn (stephan.petersenn@uni-due.de)
Vivien S Bonert (Vivien.Bonert@cshs.org)
Marco Boscaro (m.boscaro@ao-umbertoprimo.marche.it)

Version: 3 Date: 7 April 2010

Author's response to reviews:

As requested, an Author Contributions Section has been added before the acknowledgements and reference list. The added text has been approved by all authors and reads:

Author contributions

BMKB, AC, SP, VSB and MB all participated in an independent CME symposium at the 89th Annual Meeting of the Endocrine Society. This manuscript was developed directly from those presentations, which each author prepared him/herself. BMKB and SP directed the development of the manuscript, based on the content of the symposium presentations, with editorial assistance from Keri Wellington. All authors reviewed the manuscript and provided critical input. All authors read and approved the final manuscript.